Author's response to reviews

Title: Pseudo tumoral amyloidosis of the trachea revealed by a false asthma

Authors:

Mounia Serraj (moonserraj@yahoo.fr)
Imane Kamaoui (kamaouiimane@yahoo.fr)
kawtar Znati (kawtarznati@yahoo.fr)
salma Kouara (kouarasalma@gmail.com)
ferdaous sahnoune (ferdaous.05@hotmail.com)
bouchra amara (amarabouchra@gmail.com)
mohammed el biaze (melbiaze@gmail.com)
siham tizniti (sihamtizniti@yahoo.fr)
afaf amarti (amartiafaf@yahoo.fr)
mohammed chakib benjelloun (mcbenjelloun@yahoo.fr)

Version: 2  Date: 22 June 2011

Author's response to reviews:

The revisions were made based on your recommendatons. I also added another figure of bronchoscopy.